Author's response to reviews

Title: Effect of maternal calcium intake during pregnancy on children blood pressure. A systematic review of the literature.

Authors:

Eduardo E Bergel (bergel@allstat.org)
Aluisio JD Barros (abarros.epi@gmail.com)

Version: 4 Date: 24 February 2007

Author's response to reviews: see over
Dr Lolu da-Silva  
Assistant Editor, 
BMC-series journals  

ATH/rh  
24 February 2007  

Dear Dr da-Silva,  

Re: Manuscript Ref 6709610391058574 “Effect of maternal calcium intake during pregnancy on children blood pressure. A systematic review of the literature.”  

Thanks for accepting our manuscript for publication in BMC Pediatrics.  

We have included in the paper the reviewer comments.  

Many thanks for your time.  

Yours sincerely,  

Eduardo Bergel  
Department of Mother & Child Health Research,  
Institute for Clinical Effectiveness and Health Policy.  
Address: School of Public Health, School of Medicine,  
University of Buenos Aires,  
Marcelo T de Alvear 2202, (C1122AAJ)  
Buenos Aires, Argentina.
Response to Reviewer Comments

Manuscript Ref 6709610391058574 “Effect of maternal calcium intake during pregnancy on children blood pressure. A systematic review of the literature.”

Reviewer - Malinee Laopaiboon

Minor Essential Revisions

Comment 1. Please check I² in the forth line of the first paragraph of results part on page 9. It is still < 1%.
   It must be < 10%. However, I suggest to modify the sentence to be
   ‘……12 months of age (I² = 53%), and small for studies on older children (I² = 9.4%).’
   The figure of I² = 9.4% is not very small.

   We have included in the new version the modification suggested by the reviewer.

Comment 2. The author should mention ‘table 4’ somewhere in the second paragraph

   We have included in the new version the modification suggested by the reviewer.